Author's response to reviews


Authors:

Rodrick RK Kabangila (kajuna@doctor.com)
Kilonzo KS Semvua (sekipcb@yahoo.com)
Peter PR Rambau (prambau@bugando.ac.tz)
Kahima KJ Jackson (kahima61j@yahoo.com)
Stephen SEM Mshana (mshana72@yahoo.com)
Hyasinta HJ Jaka (hjaka2@hotmail.com)
Robert RNP Peck (mp2002@gmail.com)

Version: 2 Date: 15 February 2011

Author's response to reviews:

Authors were wrongly given middle names' initials. The initials were meant to be for all names of each author.